Author’s response to reviews

Title: Hepatitis C-related cryoglobulinemic neuropathy: potential role of oxcarbazepine for pain control

Authors:

Rita Moretti (moretti@units.it)
Matteo Dal Ben (mdalben@csf.units.it)
Silvia Gazzin (silvia.gazzin@csf.units.it)
Claudio Tiribelli (ctliver@fegato.it)

Version: 3 Date: 10 Jan 2018

Author’s response to reviews:

January 10, 2018

BMGE-D-17-00427-R3

Hepatitis C-related cryoglobulinemic neuropathy: Potential role of oxcarbazepine for pain control.

Rita Moretti; Paola Caruso; Matteo Dal Ben; Silvia Gazzin and Claudio Tiribelli.

BMC Gastroenterology

Dear Editor,

The last changes have been made as required:

1. Dr. Paola Caruso has been added in the editorial submission system.

2. The abstract as well the manuscript, are actually organized in: Background; Methods; Results; Conclusions, as well the manuscript in accordance to BMC Gastroenterology formatting.
3. The sections Ethics approval and consent to participate Consent for publication Availability of data and material Competing interests, Funding, Authors' contributions, and Acknowledgements have been organized after the subheading 'Declarations', and modified as request:

- The full name of the ethics committee and the institute to which it belongs have been added.

- We specify that the patients gave their written informed consent to participate to the study, as well to publish the results of the study.

- We declared that the datasets generated and/or analysed during the current study are not publicly available due confidentiality reason but are available from Dr. Rita Moretti on reasonable request.

- The rebuttal has been removed from the inventory.

- We uploaded a clean version of the manuscript not including track changes or highlighting (BMGE-D-17-00427-R3).

Sincerely,

Silvia Gazzin